\documentclass[10pt]{studiamnew}
\usepackage{graphicx}
\usepackage{amsmath}
\usepackage{amssymb}
\sloppy
%%%%%%
\allowdisplaybreaks
\usepackage{cite}
%\usepackage{refcheck}
%%%%%%%
\newtheorem{theorem}{Theorem}[section]
\newtheorem{lemma}[theorem]{Lemma}
\newtheorem{corollary}[theorem]{Corollary}
\newtheorem{proposition}[theorem]{Proposition}
\newtheorem{axiom}[theorem]{Axiom}
\newtheorem{case}[theorem]{Case}
\newtheorem{condition}[theorem]{Condition}
\newtheorem{conjecture}[theorem]{Conjecture}
\newtheorem{criterion}[theorem]{Criterion}
\newtheorem{definition}[theorem]{Definition}

\theoremstyle{definition}
\newtheorem{remark}[theorem]{Remark}
\newtheorem{algorithm}[theorem]{Algorithm}
\newtheorem{problem}[theorem]{Problem}
\newtheorem{example}[theorem]{Example}
\newtheorem{exercise}[theorem]{Exercise}
\newtheorem{solution}[theorem]{Solution}
\newtheorem{notation}[theorem]{Notation}

\renewcommand{\theequation}{\thesection.\arabic{equation}}
\numberwithin{equation}{section}

\begin{document}
\begin{thebibliography}{99}
\bibitem{ADD} Alomari, M.,  Darus, M., Dragomir, S. S., \emph{New inequalities of Hermite--Hadamard’s type for functions
whose second derivatives absolute values are quasiconvex}, Tamkang J. Math., {\bf41}(2010), 353--359.

\bibitem{ADD2} Alomari, M.,  Darus, M., Dragomir, S. S., \emph{Inequalities of Hermite–Hadamard’s type for functions whose derivatives absolute values are quasi-convex}, RGMIA Res. Rep. Collect., {\bf12}(Supplement) (2009), Article ID 14.

\bibitem{ADK} Alomari, M., Darus, M., Kirmaci, U. S., \emph{Refinements of Hadamard-type inequalities for quasiconvex functions with applications to trapezoidal formula and to special means}, Comput. Math. Appl., {\bf59} (2010), 225--232.

\bibitem{BGD} Barani, A., Ghazanfari,  A. G., Dragomir, S. S., \emph{Hermite--Hadamard inequality through prequasiinvex functions}, RGMIA Res. Rep. Coll., {\bf14}(2011), Article 48.

\bibitem{BM} Ben-Israel, A., Mond, B., \emph{What is invexity?}, J. Austral. Math. Soc. Ser. E Appl. Math., {\bf28}(1986),
1--9. 

\bibitem{CK} Chen, H., Katugampola, U. N. \emph{Hermite--Hadamard and Hermite--Hadamard--Fejér type inequalities for generalized fractional integrals}, J. Math. Anal. Appl., {\bf446}(2017), no. 2, 1274--1291.

\bibitem{DS} Rostamian Delavar , M., Dragomir, S. S., \emph{Hermite--Hadamard’s mid- point type inequalities for generalized fractional integrals}, RACSAM, {\bf114},  73 (2020).

\bibitem{DAS} Rostamian Delavar, M., Aslani,  S. M., De La Sen, M., \emph{Hermite-Hadamard-Fej\'er inequality related to generalized convex functions via fractional integrals}, J. Math., {\bf2018}(2018), Article ID 5864091, 10 pp. 

\bibitem{DP} Dragomir, S. S., Pearce, C. E. M., \emph{Quasi-convex functions and Hadamard’s inequality}, Bull. Aust. Math. Soc., {\bf57}(1998), 377--385.


\bibitem{JH} Hadamard, J., \emph{Etude sur les properties des fonctions entries et an particular d'une fonction considree par, Riemann}, J. Math. Pures. et Appl. {\bf58}(1893), 171--215.

\bibitem{HM} Hanson, M. A., Mond, B., \emph{Convex transformable programming problems and invexity}, J. Inf. Opt. Sci., {\bf8}(1987), 201--207.

\bibitem{ion} Ion, D. A., \emph{Some estimates on the Hermite-Hadamard inequality through quasi-convex functions}, Annals of University of Craiova, Math. Comp. Sci. Ser., {\bf34}(2007), 82--87.

\bibitem{II} Iscan, I., \emph{Hermite--Hadamard's inequalities for prequasiinvex functions via fractional integrals}, Konuralp J. Math., {\bf2}(2014), no. 2, 76--84.

\bibitem{UNK1} Katugampola, U. N., \emph{New approach to a generalized fractional integral}, Appl. Math. Comput., {\bf218}(2011), no. 3, 860--865.

\bibitem{UNK2} Katugampola, U. N., \emph{A new approach to generalized fractional derivatives}, Bull. Math. Anal. Appl., {\bf6}(2014), no. 4, 1--15.

\bibitem{SK1} Kermausuor, S., \emph{Generalized Ostrowski-type inequalities involving second derivatives via the Katugampola fractional integrals}, J. Nonlinear Sci. Appl., {\bf12}(2019), no. 8, 509--522.


\bibitem{SK2} Kermausuor, S., \emph{Simpson's type inequalities via the Katugampola fractional integrals for $s$-convex functions}, Kragujevac J. Math., {\bf45}(2021), no. 5, 709--720.


%\bibitem{KN1} Kermausuor, S., Nwaeze,  E. R., \emph{A note on some new Hermite--Hadamard type inequalities for functions whose $n$th derivatives are strongly $\eta$-convex},  Int. J. Nonlinear Anal. Appl., In press.

\bibitem{KN2} Kermausuor, S., Nwaeze,  E. R., \emph{Some new inequalities  involving the Katugampola fractional integrals for strongly $\eta$-convex functions}, Tbilisi Math. J., {\bf12}(2019), no. 1, 117--130.


\bibitem{KNT1} Kermausuor, S., Nwaeze, E. R., Tameru, A. M., \emph{New integral inequalities via the Katugampola fractional integrals for functions whose second derivatives are strongly $\eta$-convex}, Mathematics, {\bf7}(2019), no. 2, 183.


\bibitem{MAL} Latif, M. A., \emph{Some inequalities for differentiable prequasiinvex functions with applications}, Konulrap J. Math., {\bf1}(2013), no. 2, 17--29.

\bibitem{LD} Latif, M. A., Dragomir, S. S., \emph{Some weighted integral inequalities for differentiable preinvex and prequasiinvex functions with applications}, J. Inequal. Appl. {\bf 2013:575} (2013).

\bibitem{ERN} Nwaeze, E. R., \emph{Inequalities of the Hermite--Hadamard type for quasi-convex functions via the $(k,s)$-Riemann--Liouville fractional integrals}, Fract. Differ. Calc., \textbf{8}(2018), no. 2, 327--336.

%\bibitem{NK1} Nwaeze, E. R., Kermausuor, S., \emph{Certain results associated with the strongly $\eta$-convex function with modulus $\mu\geq 0$}, Acta Math. Univ. Comenianae, {\bf LXXXIX}(2020), no. 1, 61--74.


\bibitem{NKT} Nwaeze, E. R., Kermausuor, S., Tameru, A. M., \emph{Some new $k$-Riemann--Liouville Fractional integral inequalities associated with the strongly $\eta$-quasiconvex functions with modulus $\mu\geq0$}, J.  Inequal. Appl. {\bf2018:139} (2018).

\bibitem{pini} Pini, R., \emph{Invexity and generalized convexity}, Optimization, {\bf22}(2001), 513--525.

\bibitem{IP} Podlubny, I., \emph{Fractional differential equations: Mathematics in Science and Engineering}, Academic Press, San Diego, CA. 1999.

\bibitem{JP} Ponstein, J., \emph{Seven  types of convexity},  SIAM  Review {\bf9}  (1967),  115--119.

\bibitem{PBM} Prudnikov, A. P., Brychkov, Y. A., Marichev, O. I., \emph{Integral and series. In: Elementary Functions}, {\bf1} Nauka, Moscow, 1981.

\bibitem{RV}  Roberts, A. W., Varberg, D. E., \emph{Convex functions}, Academic  Press,  New  York,  1973.

\bibitem{SKM} Samko, S. G., Kilbas,  A. A., Marichev, O. I., \emph{Fractional Integrals and Derivatives. Theory and Applications}, Gordon and Breach, Amsterdam, 1993.

\bibitem{SM} Set, E. and Mumcu, I., \emph{Hermite--Hadamard type inequalities for quasi-convex functions via Katugampola fractional integrals}, Int. J. Anal. Appl., {\bf16}(2018), no. 4, 605--613.

\bibitem{WZZ} Wang, J., Zhu, C., Zhou, Y., \emph{New generalized Hermite–Hadamard type inequalities and applications to special means}, J. Inequal. Appl. {\bf2013: 325} (2013).

\bibitem{WM} Weir, T., Mond, B., \emph{Pre-invex functions in multiple objective optimization}, J. Math. Anal. Appl., {\bf136}(1988), no. 1, 29--38. 

\bibitem{YYT} Yang,  X. M., Yang, X. Q., Teo, K. L., \emph{Characterizations and applications of prequasiinvex
functions}, J. Optim. Theory Appl., {\bf110}(2001), no. 3, 645--668.



\end{thebibliography}

%
\setcounter{page}{1}
\setcounter{firstpage}{1}
\setcounter{lastpage}{4}
\renewcommand{\currentvolume}{??}
\renewcommand{\currentyear}{??}
\renewcommand{\currentissue}{??}
%
\title[Fractional integral inequalities for prequasiinvex functions]{New midpoint and trapezoidal-type inequalities for prequasiinvex functions via generalized fractional integrals}
\author{Seth Kermausuor}
\address{Department of Mathematics and Computer Science, \\Alabama State University,
Montgomery, AL 36101, USA}
\email{skermausour@alasu.edu}
%
\author{Eze R. Nwaeze}
\address{Department of Mathematics and Computer Science, \\Alabama State University,
Montgomery, AL 36101, USA}
\email{enwaeze@alasu.edu}
%
\subjclass{26A33, 26A51, 26D10, 26D15.}
\keywords{Hermite--Hadamard inequality, midpoint-type inequalities, trapezoidal-type inequalities, quasi-convex functions, prequasiinvex functions,  H\"older's inequality, power mean inequality, Katugampola fractional integrals, Riemann--Liouville fractional integrals, Hadamard fractional integrals.}
\begin{abstract}
In this work, we establish some new midpoint and trapezoidal type inequalities for prequasiinvex functions via the Katugampola fractional integrals. Some of the results obtained in this paper are generalizations of some earlier results in the literature. 
\end{abstract}
\maketitle

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%intro
\section{Introduction}
A function $f:[a, b]\to\mathbb{R}$ is said to be convex on $[a, b]$ if 
\begin{align}
\nonumber
f(tx+(1-t)y)\leq tf(x)+(1-t)f(y)
\end{align}
  for all $x, y\in [a, b]$ and $t\in[0,1]$ (see \cite{JP, RV}).
The following result which holds for convex functions is known in the literature as the Hermite--Hadamard inequality.
\begin{theorem}[\cite{JH}]
If $f:[a, b]\to\mathbb{R}$ is convex on $[a, b]$ with $a<b$, then 
\begin{align}
\nonumber
f\Big(\frac{a+b}{2}\Big)\leq\frac{1}{b-a}\int_a^bf(x)dx\leq\frac{f(a)+f(b)}{2}.
\end{align}
\end{theorem}

Many authors have studied and generalized the Hermite--Hadamard inequality in several ways via different classes of convex functions. For some recent results related to the Hermite--Hadamard inequality, we refer the interested reader to the papers \cite{ADD, ERN, NKT, II, MAL, LD, BGD,DP, ADK, ADD2, KN2, KNT1}.%\cite{ADD, ERN, NKT, II, MAL, LD, BGD,DP, ADK, ADD2, KN1, KN2, KNT1, NK1}.

The concept of quasi-convexity which generalizes the concept of convexity is defined as follows. 
\begin{definition}[See \cite{JP, RV}]
A function $f:[a, b]\to\mathbb{R}$ is said to be quasi-convex on $[a, b]$ if 
\begin{align}
\nonumber
f(tx+(1-t)y)\leq \max\{f(x), f(y)\}
\end{align}
  for all $x, y\in [a, b]$ and $t\in[0,1]$.
\end{definition}
In \cite{ion}, Ion introduced the following Hermite--Hadamard type inequalities also known as trapezoidal-type inequalities for quasi-convex functions as follows.
\begin{theorem}\label{ion1}
Let $f:[a, b]\to\mathbb{R}$ be a differentiable function on $(a, b)$. If $|f'|$ is quasi-convex on $[a, b]$, then the following inequality holds:
\begin{align}
\nonumber
\left|\frac{f(a)+f(b)}{2}-\frac{1}{b-a}\int_a^bf(x)dx\right|\leq\frac{b-a}{4}\max\{|f'(a)|, |f'(b)|\}.
\end{align}
\end{theorem}
\begin{theorem}\label{ion2}
Let $f:[a, b]\to\mathbb{R}$ be a differentiable function on $(a, b)$. If $\displaystyle |f'|^{\frac{p}{p-1}}$, $p>1$ is quasi-convex on $[a, b]$, then the following inequality holds:
\begin{align}
\nonumber
\left|\frac{f(a)+f(b)}{2}-\frac{1}{b-a}\int_a^bf(x)dx\right|\leq\frac{b-a}{2(p+1)^{1/p}}\left(\max\left\{|f'(a)|^{\frac{p}{p-1}}, |f'(b)|^{\frac{p}{p-1}}\right\}\right)^{\frac{p-1}{p}}.
\end{align}
\end{theorem}
For more results related to quasi-convex functions, we refer the interested reader to the papers \cite{DP, ADK, ADD, ADD2}. The concept of preinvexity was introduced in \cite{BM, HM, WM} as a generalization of convexity as follows.
\begin{definition}\label{pre}
Let $I\subseteq\mathbb{R}$ and $\eta:I\times I\to\mathbb{R}$ be a bifunction. $I$ is said to be an invex function with respect to $\eta$, if 
\begin{align}
\nonumber
x+t\eta(y, x)\in I ~\mbox{for all}~ x, y\in I ~\mbox{and}~ t\in[0, 1].
\end{align}
If $I\subseteq\mathbb{R}$ is an invex set with respect to the bifunction $\eta$, then a function $f: I\to\mathbb{R}$ is said to be a preinvex function with respect to $\eta$, if 
\begin{align}
\nonumber
f(x+t\eta(y, x))\leq (1-t) f(x)+tf(y) ~\mbox{for all}~ x, y\in I ~\mbox{and}~ t\in[0, 1].
\end{align}
\end{definition}
\begin{remark}
If $\eta(y, x)=y-x$ in Definition \ref{pre}, then we have that $f$ is a convex function. Thus, every convex function is a preinvex function with respect to the bifunction $\eta(y,x)=y-x$. However, not every preinvex function is a convex function (see \cite{WM} for more details).
\end{remark}
In a similar way, the concept of quasi-convexity has been generalized in the following definition.
\begin{definition}[\cite{pini}]\label{prequa}
If $I\subseteq\mathbb{R}$ is an invex set with respect to the bifunction $\eta$, then a function $f: I\to\mathbb{R}$ is said to be prequasiinvex with respect to $\eta$, if 
\begin{align}
\nonumber
f(x+t\eta(y, x))\leq \max\{f(x), f(y)\} ~\mbox{for all}~ x, y\in I ~\mbox{and}~ t\in[0, 1].
\end{align}
\end{definition}
\begin{remark}
Every quasi-convex function is a prequasiinvex function with respect to the bifunction $\eta(y,x)=y-x$. However, not every prequasinvex function is a quasi-convex function (see \cite{YYT} for more details).
\end{remark}
Barani et al. \cite{BGD} established the following trapezoidal-type inequalities for prequasiinvex functions which are generalizations of Theorem \ref{ion1} and Theorem \ref{ion2}.
\begin{theorem}\label{bar1}
Let $A\subseteq\mathbb{R}$ be an open invex subset with respect to $\eta: A \times A\to\mathbb{R}$. Suppose that $f : A \to\mathbb{R}$ is a differentiable function. If $|f'|$ is prequasiinvex on $A$, then for every $a, b\in A$ the following inequality holds:
\begin{align}
\nonumber
&\left|\frac{f(a)+f(a+\eta(b, a))}{2}-\frac{1}{\eta(b,a)}\int_a^{a+\eta(b, a)}f(x)dx\right|\\
\nonumber
&\quad\quad\quad\quad\quad\leq\frac{|\eta(b, a)|}{4}\max\{|f'(a)|, |f'(b)|\}.
\end{align}

\end{theorem}
\begin{theorem}\label{bar2}
Let $A\subseteq\mathbb{R}$ be an open invex subset with respect to $\eta: A \times A\to\mathbb{R}$. Suppose that $f : A \to\mathbb{R}$ is a differentiable function. If $\displaystyle |f'|^{\frac{p}{p-1}}$ is prequasiinvex on $A$, then for every $a, b\in A$ the following inequality holds:
\begin{align}
\nonumber
&\left|\frac{f(a)+f(a+\eta(b, a))}{2}-\frac{1}{\eta(b,a)}\int_a^{a+\eta(b, a)}f(x)dx\right|\\
\nonumber
&\quad\quad\quad\quad\quad\quad~~~~~~~~~~\leq\frac{|\eta(b, a)|}{2(p+1)^{1/p}}\left(\max\left\{|f'(a)|^{\frac{p}{p-1}}, |f'(b)|^{\frac{p}{p-1}}\right\}\right)^{\frac{p-1}{p}}.
\end{align}

\end{theorem}
For more information and results related to prequasiinvex functions, we refer the interested reader to the papers \cite{pini, YYT, II, MAL, LD}. In \cite{II}, the author generalized Theorem \ref{bar1} and Theorem \ref{bar2} using the Riemann--Liouville fractional integrals.

Our goal in this paper is to provide some midpoint and trapizoidal type inequalities for functions whose derivative in absolute value to some exponents are prequasiinvex via the Katugampola fractional integrals. Some of our results generalize the results in \cite{II}. We end this section with the definitions of the Riemann--Liouville, Hadamard and Katugampola fractional integrals and some preliminary results.

\begin{definition}[\cite{IP}]
 The left- and right-sided Riemann--Liouville fractional integrals of order $\alpha > 0$ of $f$ are defined by
\begin{align}
\nonumber
J_{a+}^\alpha f(x):=\frac{1}{\Gamma(\alpha)}\int_a^x(x-t)^{\alpha-1}f(t)dt
\end{align}
and 
\begin{align}
\nonumber
J_{b-}^\alpha f(x):=\frac{1}{\Gamma(\alpha)}\int_x^b(t-x)^{\alpha-1}f(t)dt
\end{align}
with $a<x<b$  and  $\Gamma(\cdot)$ is the gamma function given by 
\begin{align}
\nonumber
\Gamma(x):=\int_0^\infty t^{x-1}e^{-t}dt, ~~~~Re(x)>0
\end{align}
with the property that $\Gamma(x+1)=x\Gamma(x)$.
\end{definition}

\begin{definition}[\cite{SKM}]
 The left- and right-sided Hadamard fractional integrals of order $\alpha > 0$ of $f$ are defined by
\begin{align}
\nonumber
H_{a+}^\alpha f(x):=\frac{1}{\Gamma(\alpha)}\int_a^x\left(\ln\frac{x}{t}\right)^{\alpha-1}\frac{f(t)}{t}dt
\end{align}
and 
\begin{align}
\nonumber
H_{b-}^\alpha f(x):=\frac{1}{\Gamma(\alpha)}\int_x^b\left(\ln\frac{t}{x}\right)^{\alpha-1}\frac{f(t)}{t}dt.
\end{align}
\end{definition}
\begin{definition}
 $X_c^p(a, b)~~ (c\in\mathbb{R},~1\leq p\leq\infty)$ denotes the space of all complex-valued Lebesgue measurable functions $f$ for which $\|f\|_{X_c^p}<\infty$, where the norm $\|\cdot\|_{X_c^p}$ is defined by 
\begin{align}
\nonumber
\|f\|_{X_c^p}=\left(\int_a^b|t^cf(t)|^p\frac{dt}{t}\right)^{1/p} ~~~(1\leq p<\infty)
\end{align}
and for $p=\infty$
\begin{align}
\nonumber
\|f\|_{X_c^\infty}=\mbox{ess}\sup_{a\leq t\leq b}|t^cf(t)|.
\end{align}
\end{definition}
In 2011, Katugampola \cite{UNK1} introduced a new fractional integral operator which generalizes the Riemann--Liouville and Hadamard fractional integrals as follows:
\begin{definition}\label{1DK}
Let $[a, b]\subset\mathbb{R}$ be a finite interval. Then, the left- and right-sided Katugampola
fractional integrals of order $\alpha > 0$ of $f\in X^p_c (a, b)$ are defined by
\begin{align}
\nonumber
{}^\rho I_{a+}^\alpha f(x):=\frac{\rho^{1-\alpha}}{\Gamma(\alpha)}\int_a^x\frac{t^{\rho-1}}{(x^\rho-t^\rho)^{1-\alpha}}f(t)dt
\end{align}
and 
\begin{align}
\nonumber
{}^\rho I_{b-}^\alpha f(x):=\frac{\rho^{1-\alpha}}{\Gamma(\alpha)}\int_x^b\frac{t^{\rho-1}}{(t^\rho-x^\rho)^{1-\alpha}}f(t)dt
\end{align}
with $a<x<b$ and $\rho>0$, if the integrals exist.
\end{definition}
\begin{remark}
It is shown in \cite{UNK1} that the Katugampola fractional integral operators are well-defined on $ X^p_c (a, b)$.
\end{remark}
\begin{theorem}[\cite{UNK1}]
Let $\alpha>0$ and $\rho>0$. Then for $x>a$
\begin{enumerate}
\item $\displaystyle\lim_{\rho\to1}{}^\rho I_{a+}^\alpha f(x)=J_{a+}^\alpha f(x)$,
\item $\displaystyle\lim_{\rho\to0^+}{}^\rho I_{a+}^\alpha f(x)=H_{a+}^\alpha f(x)$.
\end{enumerate}
Similar results also hold for right-sided operators.
\end{theorem}
For more information about the Katugampola fractional integrals and related results, we refer the interested reader to the papers \cite{CK, UNK1, UNK2, SK1, SK2}.

\begin{lemma}[See \cite{PBM, WZZ}]\label{SO}
For any $\alpha\in[0, 1]$ and $x, y\in[0, 1]$, we have 
\begin{align}
\nonumber
|x^\alpha-y^\alpha|\leq|x-y|^\alpha.
\end{align}
\end{lemma}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%mainresult
\section{Main Results}
\subsection{Midpoint-type inequalities }
The following lemma is a generalization of \cite[Lemma 16]{DAS} via the Katugampola fractional integrals. 

%=============================================================================mlem
\begin{lemma}\label{mlem}
Let $\alpha, \rho> 0$, $I\subseteq\mathbb{R}$ be an open invex set with respect to the bifunction $\eta: I\times I\to\mathbb{R}$ and $ f : I\to\mathbb{R}$ be a differentiable mapping on $I$. If $a, b>0$ with $a<b$ such that $a^\rho, b^\rho\in I$, $\eta(b^\rho, a^\rho)>0$ and $f'\in L_1\Big(\left[a^\rho, a^\rho+\eta(b^\rho, a^\rho)\right]\Big)$, then the following equality via the fractional integrals holds:
\begin{align}\label{mlemID}
\nonumber
f&\left(\frac{2a^\rho+\eta(b^\rho,a^\rho)}{2}\right)-\frac{\rho^{\alpha}\Gamma(\alpha+1)}{2\eta(b^\rho,a^\rho)^{\alpha}}\bigg[{}^\rho I_{a+}^\alpha f(a^\rho+\eta(b^\rho,a^\rho))\\
\nonumber
&\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad+{}^\rho I_{\left(\sqrt[\rho]{a^\rho+\eta(b^\rho,a^\rho)}\right)-}^\alpha f(a^\rho)\bigg]\\
&=\frac{\eta(b^\rho,a^\rho)\rho}{2}\Big(I_1+I_2+I_3+I_4\Big),
\end{align}
where
\begin{align}
\nonumber
&I_1=\int_0^{\sqrt[\rho]{1/2}}t^{(\alpha+1)\rho-1}f'(a^\rho+t^\rho\eta(b^\rho, a^\rho))dt,\\
\nonumber
&I_2=-\int_0^{\sqrt[\rho]{1/2}}t^{(\alpha+1)\rho-1}f'(a^\rho+(1-t^\rho)\eta(b^\rho, a^\rho))dt,\\
\nonumber
&I_3=\int_{\sqrt[\rho]{1/2}}^1(t^{\alpha\rho}-1)t^{\rho-1}f'(a^\rho+t^\rho\eta(b^\rho, a^\rho))dt
\end{align}
and
\begin{align}
\nonumber
I_4=\int_{\sqrt[\rho]{1/2}}^1(1-t^{\alpha\rho})t^{\rho-1}f'(a^\rho+(1-t^\rho)\eta(b^\rho, a^\rho))dt.
\end{align}

\end{lemma}

\begin{proof}
By integrating by parts, we have 
\begin{align}\label{A1}
\nonumber
I_1&=\int_0^{\sqrt[\rho]{1/2}}t^{(\alpha+1)\rho-1}f'(a^\rho+t^\rho\eta(b^\rho, a^\rho))dt\\
\nonumber
&=\frac{t^{\alpha\rho}}{(b^\rho-a^\rho)\rho}f(a^\rho+t^\rho\eta(b^\rho, a^\rho))\bigg|_0^{\sqrt[\rho]{1/2}}\\
\nonumber
&\quad\quad\quad\quad-\frac{\alpha}{\eta(b^\rho,a^\rho)}\int_0^{\sqrt[\rho]{1/2}}t^{\alpha\rho-1}f(a^\rho+t^\rho\eta(b^\rho, a^\rho))dt\\
\nonumber
&=\frac{2^{-\alpha}}{\eta(b^\rho,a^\rho)\rho}f\left(\frac{2a^\rho+\eta(b^\rho,a^\rho)}{2}\right)\\
&\quad\quad\quad\quad-\frac{\alpha}{\eta(b^\rho, a^\rho)}\int_0^{\sqrt[\rho]{1/2}}t^{\alpha\rho-1}f(a^\rho+t^\rho\eta(b^\rho, a^\rho))dt.
\end{align}
Similarly, we have 
\begin{align}\label{A2}
\nonumber
I_2&=\frac{2^{-\alpha}}{\eta(b^\rho,a^\rho)\rho}f\left(\frac{2a^\rho+\eta(b^\rho,a^\rho)}{2}\right)\\
&\quad\quad\quad\quad-\frac{\alpha}{\eta(b^\rho, a^\rho)}\int_0^{\sqrt[\rho]{1/2}}t^{\alpha\rho-1}f(a^\rho+(1-t^\rho)\eta(b^\rho, a^\rho))dt,
\end{align}
\begin{align}\label{A3}
\nonumber
I_3&=\int_{\sqrt[\rho]{1/2}}^1(t^{\alpha\rho}-1)t^{\rho-1}f'(a^\rho+t^\rho\eta(b^\rho, a^\rho))dt\\
\nonumber
&=\frac{t^{\alpha\rho}-1}{\eta(b^\rho,a^\rho)\rho}f(a^\rho+t^\rho\eta(b^\rho, a^\rho))\bigg|_{\sqrt[\rho]{1/2}}^1\\
\nonumber
&\quad\quad\quad\quad-\frac{\alpha}{\eta(b^\rho,a^\rho)}\int_{\sqrt[\rho]{1/2}}^1t^{\alpha\rho-1}f(a^\rho+t^\rho\eta(b^\rho, a^\rho))dt\\
\nonumber
&=\frac{1-2^{-\alpha}}{\eta(b^\rho,a^\rho)\rho}f\left(\frac{2a^\rho+\eta(b^\rho,a^\rho)}{2}\right)\\
&\quad\quad\quad\quad-\frac{\alpha}{\eta(b^\rho,a^\rho)}\int_{\sqrt[\rho]{1/2}}^1t^{\alpha\rho-1}f(a^\rho+t^\rho\eta(b^\rho, a^\rho))dt
\end{align}
and 
\begin{align}\label{A4}
\nonumber
I_4&=\frac{1-2^{-\alpha}}{\eta(b^\rho,a^\rho)\rho}f\left(\frac{2a^\rho+\eta(b^\rho,a^\rho)}{2}\right)\\
&\quad\quad\quad\quad-\frac{\alpha}{\eta(b^\rho,a^\rho)}\int_{\sqrt[\rho]{1/2}}^1t^{\alpha\rho-1}f(a^\rho+(1-t^\rho)\eta(b^\rho, a^\rho))dt.
\end{align}
Now, by using \eqref{A1}, \eqref{A2}, \eqref{A3} and \eqref{A4}, we have
\begin{align}\label{A5}
\nonumber
\frac{2}{\eta(b^\rho,a^\rho)\rho}f\left(\frac{2a^\rho+\eta(b^\rho,a^\rho)}{2}\right)&-\frac{\alpha}{\eta(b^\rho,a^\rho)}\bigg[\int_0^1t^{\alpha\rho-1}f(a^\rho+t^\rho\eta(b^\rho,a^\rho))dt\\
\nonumber
&+\int_0^1t^{\alpha\rho-1}f(a^\rho+(1-t^\rho)\eta(b^\rho, a^\rho))dt\bigg]\\
&=I_1+I_2+I_3+I_4.
\end{align}
By using change of variables and Definition \ref{1DK}, we have
\begin{align}\label{A6}
\int_0^1t^{\alpha\rho-1}f(a^\rho+t^\rho\eta(b^\rho,a^\rho))dt=\frac{\rho^{\alpha-1}\Gamma(\alpha)}{\eta(b^\rho,a^\rho)^\alpha}{}^\rho I_{\left(\sqrt[\rho]{a^\rho+\eta(b^\rho,a^\rho)}\right)-}^\alpha f(a^\rho)
\end{align}
and 
\begin{align}\label{A7}
\int_0^1t^{\alpha\rho-1}f(a^\rho+(1-t^\rho)\eta(b^\rho,a^\rho))dt=\frac{\rho^{\alpha-1}\Gamma(\alpha)}{\eta(b^\rho,a^\rho)^\alpha}{}^\rho I_{a+}^\alpha f(a^\rho+\eta(b^\rho,a^\rho)).
\end{align}
Substituting \eqref{A6} and \eqref{A7} in \eqref{A5}, we obtain
\begin{align}\label{A8}
\nonumber
I_1+I_2+I_3+I_4&=\frac{2}{\eta(b^\rho,a^\rho)\rho}f\left(\frac{2a^\rho+\eta(b^\rho,a^\rho)}{2}\right)-\frac{\rho^{\alpha-1}\Gamma(\alpha+1)}{\eta(b^\rho,a^\rho)^{\alpha+1}}\\
&\quad\quad\times\bigg[{}^\rho I_{a+}^\alpha f(a^\rho+\eta(b^\rho,a^\rho))
+{}^\rho I_{\left(\sqrt[\rho]{a^\rho+\eta(b^\rho,a^\rho)}\right)-}^\alpha f(a^\rho)\bigg].
\end{align}
The desired identity in \eqref{mlemID} follows from \eqref{A8}. Hence, the proof is complete.
\end{proof}
\begin{remark}
If we choose $\rho=1$ in Lemma \ref{mlem}, then we obtain \cite[Lemma 16]{DAS}. Also, if $\rho\neq1$ and $\eta(x, y)=x-y$ in Lemma \ref{mlem}, then we obtain \cite[Lemma 2.1]{DS}  with a minor mistake in the identities obtained in \cite{DS} where $\Gamma(\alpha+1)$ should have been $\Gamma(\alpha)$ instead.
\end{remark}
%%%%%%%%%%%%MR1
\begin{theorem}\label{MR1}
Under the conditions of Lemma \ref{mlem}, if $|f'|^q, q\geq1$ is prequasiinvex on $I$, then the following inequality holds:
\begin{align}
\nonumber
\bigg|&f\left(\frac{2a^\rho+\eta(b^\rho,a^\rho)}{2}\right)-\frac{\rho^{\alpha}\Gamma(\alpha+1)}{2\eta(b^\rho,a^\rho)^{\alpha}}\bigg[{}^\rho I_{a+}^\alpha f(a^\rho+\eta(b^\rho,a^\rho))\\
\nonumber
&\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad+{}^\rho I_{\left(\sqrt[\rho]{a^\rho+\eta(b^\rho,a^\rho)}\right)-}^\alpha f(a^\rho)\bigg]\bigg|\\
\nonumber
&\leq\eta(b^\rho,a^\rho)\Big(\frac{1}{2}-\frac{1}{\alpha+1}+\frac{1}{2^{\alpha}(\alpha+1)}\Big)\left(\max\Big\{|f'(a^\rho)|^q, |f'(b^\rho)|^q\Big\}\right)^{1/q}.
\end{align}
\end{theorem}
\begin{proof}
By using Lemma \ref{mlem} and the properties of the absolute value, we have
\begin{align}\label{C0}
\nonumber
\bigg|&f\left(\frac{2a^\rho+\eta(b^\rho,a^\rho)}{2}\right)-\frac{\rho^{\alpha}\Gamma(\alpha+1)}{2\eta_1(b^\rho,a^\rho)^{\alpha}}\bigg[{}^\rho I_{a+}^\alpha f(a^\rho+\eta(b^\rho,a^\rho))\\
\nonumber
&\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad+{}^\rho I_{\left(\sqrt[\rho]{a^\rho+\eta(b^\rho,a^\rho)}\right)-}^\alpha f(a^\rho)\bigg]\bigg|\\
&\leq\frac{\eta(b^\rho,a^\rho)\rho}{2}\Big(|I_1|+|I_2|+|I_3|+|I_4|\Big).
\end{align}
By using the power mean inequality, we have
\begin{align}\label{C1}
|I_1|&\leq\left(\int_0^{\sqrt[\rho]{1/2}}t^{(\alpha+1)\rho-1}dt\right)^{1-1/q}\left(\int_0^{\sqrt[\rho]{1/2}}t^{(\alpha+1)\rho-1}|f'(a^\rho+t^\rho\eta(b^\rho, a^\rho))|^qdt\right)^{1/q}.
\end{align}
Using the prequasiinvexity of $|f'|^q$, we have
\begin{align}\label{C2}
|f'(a^\rho+t^\rho\eta(b^\rho, a^\rho))|^q\leq\max\Big\{|f'(a^\rho)|^q, |f'(b^\rho)|^q\Big\}.
\end{align}
Substituting \eqref{C2} in \eqref{C1}, we obtain
\begin{align}\label{C3}
|I_1|&\leq\frac{1}{2^{\alpha+1}(\alpha+1)\rho}\left(\max\Big\{|f'(a^\rho)|^q, |f'(b^\rho)|^q\Big\}\right)^{1/q}.
\end{align}
Using similar arguments, we deduce that
\begin{align}\label{C4}
|I_2|&\leq\frac{1}{2^{\alpha+1}(\alpha+1)\rho}\left(\max\Big\{|f'(a^\rho)|^q, |f'(b^\rho)|^q\Big\}\right)^{1/q},
\end{align}
\begin{align}\label{C5}
\nonumber
|I_3|&\leq\int_{\sqrt[\rho]{1/2}}^1|t^{\alpha\rho}-1|t^{\rho-1}dt\left(\max\Big\{|f'(a^\rho)|^q, |f'(b^\rho)|^q\Big\}\right)^{1/q}\\
\nonumber
&=\frac{1}{\rho}\int_{1/2}^1(1-u^{\alpha})du\left(\max\Big\{|f'(a^\rho)|^q, |f'(b^\rho)|^q\Big)\Big\}\right)^{1/q}\\
&=\frac{1}{\rho}\Big(\frac{1}{2}-\frac{1}{\alpha+1}+\frac{1}{2^{\alpha+1}(\alpha+1)}\Big)\left(\max\Big\{|f'(a^\rho)|^q, |f'(b^\rho)|^q\Big\}\right)^{1/q}
\end{align}
and
\begin{align}\label{C6}
|I_4|&\leq\frac{1}{\rho}\Big(\frac{1}{2}-\frac{1}{\alpha+1}+\frac{1}{2^{\alpha+1}(\alpha+1)}\Big)\left(\max\Big\{|f'(a^\rho)|^q, |f'(b^\rho)|^q\Big)\Big\}\right)^{1/q}.
\end{align}
The desired inequality follows from \eqref{C0} by using \eqref{C1}-\eqref{C2}.
\end{proof}
%%%%%corthm1
\begin{corollary}\label{corthm1}
If in Theorem \ref{MR1} we take $\eta(x, y)=x-y$ for all $x, y\in I$, i.e, $|f'|^q, q\geq1$, is quasiconvex, then the following inequality holds:
\begin{align}
\nonumber
\bigg|&f\left(\frac{a^\rho+b^\rho}{2}\right)-\frac{\rho^{\alpha}\Gamma(\alpha+1)}{2(b^\rho-a^\rho)^{\alpha}}\bigg[{}^\rho I_{a+}^\alpha f(b^\rho)+{}^\rho I_{b-}^\alpha f(a^\rho)\bigg]\bigg|\\
\nonumber
&\leq (b^\rho-a^\rho)\Big(\frac{1}{2}-\frac{1}{\alpha+1}+\frac{1}{2^{\alpha}(\alpha+1)}\Big)\left(\max\Big\{|f'(a^\rho)|^q, |f'(b^\rho)|^q\Big\}\right)^{1/q}.
\end{align}
\end{corollary}
\begin{remark}
It is worth noting that in \cite[Theorem 2.8]{DS} the authors established another estimate for the left hand side of the inequality in Corollary \ref{corthm1} under the condition that $|f'|$ is convex. On the other hand, since every convex function is quasiconvex it follows that the inequality in Corollary \ref{corthm1} holds if $|f'|^q, q\geq1$ is convex.
\end{remark}
%%%%%%%MR2
\begin{theorem}\label{MR2}
Under the conditions of Lemma \ref{mlem}, if $|f'|^q, q>1$ is prequasiinvex on $I$, then the following inequality holds:
\begin{align}\label{MR2ineq}
\nonumber
\bigg|&f\left(\frac{2a^\rho+\eta(b^\rho,a^\rho)}{2}\right)-\frac{\rho^{\alpha}\Gamma(\alpha+1)}{2\eta(b^\rho,a^\rho)^{\alpha}}\bigg[{}^\rho I_{a+}^\alpha f(a^\rho+\eta(b^\rho,a^\rho))\\
\nonumber
&\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad+{}^\rho I_{\left(\sqrt[\rho]{a^\rho+\eta(b^\rho,a^\rho)}\right)-}^\alpha f(a^\rho)\bigg]\bigg|\\
\nonumber
&\leq\frac{\eta(b^\rho,a^\rho)}{2}\bigg[\left(\frac{1}{2^{\alpha r}(\alpha r+1)}\right)^{1/r}+\left(2\int_{1/2}^1|u^{\alpha}-1|^rdu\right)^{1/r}\bigg]\\
&\quad\quad\times\left(\max\Big\{|f'(a^\rho)|^q, |f'(b^\rho)|^q\Big\}\right)^{1/q},
\end{align}
where $\displaystyle\frac{1}{r}+\frac{1}{q}=1$. In addition, if $\alpha\in(0, 1]$, then we have the inequality
\begin{align}\label{MR2ineq2}
\nonumber
\bigg|&f\left(\frac{2a^\rho+\eta(b^\rho,a^\rho)}{2}\right)-\frac{\rho^{\alpha}\Gamma(\alpha+1)}{2\eta(b^\rho,a^\rho)^{\alpha}}\bigg[{}^\rho I_{a+}^\alpha f(a^\rho+\eta(b^\rho,a^\rho))\\
\nonumber
&\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad+{}^\rho I_{\left(\sqrt[\rho]{a^\rho+\eta(b^\rho,a^\rho)}\right)-}^\alpha f(a^\rho)\bigg]\bigg|\\
&\leq\eta(b^\rho,a^\rho)\left(\frac{1}{2^{\alpha r}(\alpha r+1)}\right)^{1/r}\left(\max\Big\{|f'(a^\rho)|^q, |f'(b^\rho)|^q\Big\}\right)^{1/q}.
\end{align}

\end{theorem}
\begin{proof}
By using Lemma \ref{mlem} and the properties of the absolute value, we have
\begin{align}\label{D0}
\nonumber
\bigg|&f\left(\frac{2a^\rho+\eta(b^\rho,a^\rho)}{2}\right)-\frac{\rho^{\alpha}\Gamma(\alpha+1)}{2\eta(b^\rho,a^\rho)^{\alpha}}\bigg[{}^\rho I_{a+}^\alpha f(a^\rho+\eta(b^\rho,a^\rho))\\
\nonumber
&\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad+{}^\rho I_{\left(\sqrt[\rho]{a^\rho+\eta(b^\rho,a^\rho)}\right)-}^\alpha f(a^\rho)\bigg]\bigg|\\
&\leq\frac{\eta(b^\rho,a^\rho)\rho}{2}\Big(|I_1|+|I_2|+|I_3|+|I_4|\Big).
\end{align}
By using the H\"older's inequality, we have
\begin{align}\label{D1}
|I_1|&\leq\left(\int_0^{\sqrt[\rho]{1/2}}t^{\alpha\rho r}t^{\rho-1}dt\right)^{1/r}\left(\int_0^{\sqrt[\rho]{1/2}}t^{\rho-1}|f'(a^\rho+t^\rho\eta(b^\rho, a^\rho))|^qdt\right)^{1/q}.
\end{align}
Using the prequasiinvexity of $|f'|^q$, we have
\begin{align}\label{D2}
|f'(a^\rho+t^\rho\eta(b^\rho, a^\rho))|^q\leq\max\Big\{|f'(a^\rho)|^q, |f'(b^\rho)|^q\Big\}.
\end{align}
Substituting \eqref{D2} in \eqref{D1}, we obtain
\begin{align}\label{D3}
\nonumber
|I_1|&\leq\left(\frac{1}{2^{\alpha r+1}(\alpha r+1)\rho}\right)^{1/r}\left(\frac{1}{2\rho}\max\Big\{|f'(a^\rho)|^q, |f'(b^\rho)|^q\Big\}\right)^{1/q}\\
&=\frac{1}{2\rho}\left(\frac{1}{2^{\alpha r}(\alpha r+1)}\right)^{1/r}\left(\max\Big\{|f'(a^\rho)|^q, |f'(b^\rho)|^q\Big\}\right)^{1/q}.
\end{align}
Using similar arguments, we deduce that
\begin{align}\label{D4}
|I_2|&\leq\frac{1}{2\rho}\left(\frac{1}{2^{\alpha r}(\alpha r+1)}\right)^{1/r}\left(\max\Big\{|f'(a^\rho)|^q, |f'(b^\rho)|^q\Big\}\right)^{1/q},
\end{align}
\begin{align}\label{D5}
\nonumber
|I_3|&\leq\left(\int_{\sqrt[\rho]{1/2}}^1|t^{\alpha\rho}-1|^rt^{\rho-1}dt\right)^{1/r}\left(\max\Big\{|f'(a^\rho)|^q, |f'(b^\rho)|^q\Big\}\int_{\sqrt[\rho]{1/2}}^1t^{\rho-1}dt\right)^{1/q}\\
\nonumber
&=\left(\frac{1}{\rho}\int_{1/2}^1|u^{\alpha}-1|^rdu\right)^{1/r}\left(\frac{1}{2\rho}\max\Big\{|f'(a^\rho)|^q, |f'(b^\rho)|^q\Big\}\right)^{1/q}\\
&=\frac{1}{2\rho}\left(2\int_{1/2}^1|u^{\alpha}-1|^rdu\right)^{1/r}\left(\max\Big\{|f'(a^\rho)|^q, |f'(b^\rho)|^q\Big\}\right)^{1/q}
\end{align}
and
\begin{align}\label{D6}
|I_4|&\leq\frac{1}{2\rho}\left(2\int_{1/2}^1|u^{\alpha}-1|^rdu\right)^{1/r}\left(\max\Big\{|f'(a^\rho)|^q, |f'(b^\rho)|^q\Big\}\right)^{1/q}.
\end{align}
The inequality in \eqref{MR2ineq} follows from \eqref{D0} by using \eqref{D1}-\eqref{D2}. Now, if $\alpha\in(0, 1]$, then it follows from Lemma \ref{SO} that
\begin{align}\label{D7}
\int_{1/2}^1|u^{\alpha}-1|^rdu\leq\int_{1/2}^1(1-u)^{\alpha r}du=\frac{1}{2^{\alpha r+1}(\alpha r+1)}.
\end{align}
The inequality in \eqref{MR2ineq2} follows from \eqref{MR2ineq} by using \eqref{D7}. Hence, the proof is complete.
\end{proof}
%%%%%%%%%%%%%%%%corthm2
\begin{corollary}
If in Theorem \ref{MR2} we take $\eta(x, y)=x-y$ for all $x, y\in I$, i.e, $|f'|^q, q>1$, is quasiconvex, then the following inequality holds:
\begin{align}
\nonumber
\bigg|&f\left(\frac{a^\rho+b^\rho}{2}\right)-\frac{\rho^{\alpha}\Gamma(\alpha+1)}{2(b^\rho-a^\rho)^{\alpha}}\bigg[{}^\rho I_{a+}^\alpha f(b^\rho)+{}^\rho I_{b-}^\alpha f(a^\rho)\bigg]\bigg|\\
\nonumber
&\leq\frac{b^\rho-a^\rho}{2}\bigg[\left(\frac{1}{2^{\alpha r}(\alpha r+1)}\right)^{1/r}+\left(2\int_{1/2}^1|u^{\alpha}-1|^rdu\right)^{1/r}\bigg]\\
\nonumber
&\quad\quad\times\left(\max\Big\{|f'(a^\rho)|^q, |f'(b^\rho)|^q\Big\}\right)^{1/q},
\end{align}
where $\displaystyle\frac{1}{r}+\frac{1}{q}=1$. In addition, if $\alpha\in(0, 1]$, then we have the inequality
\begin{align}
\nonumber
\bigg|&f\left(\frac{a^\rho+b^\rho}{2}\right)-\frac{\rho^{\alpha}\Gamma(\alpha+1)}{2(b^\rho-a^\rho)^{\alpha}}\bigg[{}^\rho I_{a+}^\alpha f(b^\rho)+{}^\rho I_{b-}^\alpha f(a^\rho)\bigg]\bigg|\\
\nonumber
&\leq(b^\rho-a^\rho)\left(\frac{1}{2^{\alpha r}(\alpha r+1)}\right)^{1/r}\left(\max\Big\{|f'(a^\rho)|^q, |f'(b^\rho)|^q\Big\}\right)^{1/q}.
\end{align}

\end{corollary}

%%%%%%%%%%%%%%%%part2
\subsection{Trapezoidal-type inequalities}

%%%%%%mlem2
The following lemma is a generalization of Lemma 2.4 in \cite{CK} for the invex case.
\begin{lemma}\label{mlem2}
Let $\alpha, \rho>0$, $I\subseteq\mathbb{R}$ be an open invex set with respect to the bifunction $\eta: I\times I\to\mathbb{R}$ and $ f : I\to\mathbb{R}$ be a differentiable mapping on $I$. If $a, b>0$ with $a<b$ such that $a^\rho, b^\rho\in I$, $\eta(b^\rho, a^\rho)>0$ and $f'\in L_1\Big(\left[a^\rho, a^\rho+\eta(b^\rho, a^\rho)\right]\Big)$, then the following equality via the fractional integrals holds:
\begin{align}\label{mlemID2}
\nonumber
&\frac{f( a^\rho)+f( a^\rho+\eta(b^\rho, a^\rho))}{2}-\frac{\rho^{\alpha}\Gamma(\alpha+1)}{2\eta(b^\rho,a^\rho)^{\alpha}}\bigg[{}^\rho I_{\left(\sqrt[\rho]{a^\rho+\eta(b^\rho,a^\rho)}\right)-}^\alpha f(a^\rho)\\
\nonumber
&\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad+{}^\rho I_{a+}^\alpha f(a^\rho+\eta(b^\rho,a^\rho))\bigg]\\
&=\frac{\eta(b^\rho,a^\rho)\rho}{2}\int_0^1[(1-t^\rho)^\alpha-t^{\rho\alpha}]t^{\rho-1}f'( a^\rho+(1-t^\rho)\eta(b^\rho, a^\rho))dt.
\end{align}
\end{lemma}
\begin{proof}
We observe that 
\begin{align}
\nonumber
\int_0^1[(1-t^\rho)^\alpha-t^{\rho\alpha}]t^{\rho-1}f'( a^\rho+(1-t^\rho)\eta(b^\rho, a^\rho))dt=I_1-I_2,
\end{align}
where
\begin{align}
\nonumber
I_1&=\int_0^1(1-t^\rho)^{\alpha}t^{\rho-1}f'( a^\rho+(1-t^\rho)\eta(b^\rho, a^\rho))dt
\end{align} 
and 
\begin{align}
\nonumber
I_2&=\int_0^1t^{\alpha\rho}t^{\rho-1}f'( a^\rho+(1-t^\rho)\eta(b^\rho, a^\rho))dt.
\end{align}
By integrating by parts and change of variables, we have
\begin{align}\label{Q0}
\nonumber
I_1&=\int_0^1(1-t^\rho)^{\alpha}t^{\rho-1}f'( a^\rho+(1-t^\rho)\eta(b^\rho, a^\rho))dt\\
\nonumber
&=-\frac{(1-t^\rho)^{\alpha}}{\eta(b^\rho,a^\rho)\rho}f( a^\rho+(1-t^\rho)\eta(b^\rho, a^\rho))\bigg|_0^1\\
\nonumber
&\quad\quad-\frac{\alpha}{\eta(b^\rho,a^\rho)}\int_0^1(1-t^\rho)^{\alpha-1}t^{\rho-1}f( a^\rho+(1-t^\rho)\eta(b^\rho, a^\rho))dt\\
\nonumber
&=\frac{1}{\eta(b^\rho,a^\rho)\rho}f( a^\rho+\eta(b^\rho, a^\rho))\\
\nonumber
&\quad\quad-\frac{\alpha}{\eta(b^\rho,a^\rho)}\int_0^1(1-t^\rho)^{\alpha-1}t^{\rho-1}f( a^\rho+(1-t^\rho)\eta(b^\rho, a^\rho))dt\\
\nonumber
&=\frac{1}{\eta(b^\rho,a^\rho)\rho}f( a^\rho+\eta(b^\rho, a^\rho))\\
&\quad\quad-\frac{\alpha}{\eta(b^\rho,a^\rho)^{\alpha+1}}\int\limits_{a}^{\sqrt[\rho]{a^\rho+\eta(b^\rho,a^\rho)}}(u^\rho-a^\rho)^{\alpha-1}u^{\rho-1}f(u^\rho)du.
\end{align}
By using Definition \ref{1DK} and \eqref{Q0}, we have
\begin{align}\label{Q1}
I_1&=\frac{f( a^\rho+\eta(b^\rho, a^\rho))}{\eta(b^\rho,a^\rho)\rho}-\frac{\rho^{\alpha-1}\Gamma(\alpha+1)}{\eta(b^\rho,a^\rho)^{\alpha+1}}{}^\rho I_{\left(\sqrt[\rho]{a^\rho+\eta(b^\rho,a^\rho)}\right)-}^\alpha f(a^\rho).
\end{align}
By a similar argument, we have
\begin{align}\label{Q2}
I_2&=-\frac{f(a^\rho)}{\eta(b^\rho,a^\rho)\rho}+\frac{\rho^{\alpha-1}\Gamma(\alpha+1)}{\eta(b^\rho,a^\rho)^{\alpha+1}}{}^\rho I_{a+}^\alpha f(a^\rho+\eta(b^\rho,a^\rho)).
\end{align}
By using \eqref{Q1} and \eqref{Q2}, we have
\begin{align}\label{Q3}
\nonumber
I_1-I_2&=\frac{f( a^\rho)+f( a^\rho+\eta(b^\rho, a^\rho))}{\eta(b^\rho,a^\rho)\rho}-\frac{\rho^{\alpha-1}\Gamma(\alpha+1)}{\eta(b^\rho,a^\rho)^{\alpha+1}}\bigg[{}^\rho I_{\left(\sqrt[\rho]{a^\rho+\eta(b^\rho,a^\rho)}\right)-}^\alpha f(a^\rho)\\
&\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad+{}^\rho I_{a+}^\alpha f(a^\rho+\eta(b^\rho,a^\rho))\bigg].
\end{align}

The desired identity in \eqref{mlemID2} follows from \eqref{Q3}.
\end{proof}
\begin{remark}
If $\eta(x,y)=x-y$ in Lemma \ref{mlem2}, then we obtain \cite[Lemma 2.4]{CK}  with minor mistakes in the identity obtained in \cite{CK} where $\Gamma(\alpha+1)$ should have been $\Gamma(\alpha)$ and $\displaystyle\frac{b^\rho-a^\rho}{2}$ should have been $\displaystyle\frac{(b^\rho-a^\rho)\rho}{2}$  instead.

\end{remark}
%%%%%%%%%%%%%%%MR3
\begin{theorem}\label{MR3}
Under the conditions of Lemma \ref{mlem2}, if $|f'|^q, q\geq1$ is prequasiinvex on $I$, then the following inequality holds:
\begin{align}\label{MR3ineq}
\nonumber
&\bigg|\frac{f( a^\rho)+f( a^\rho+\eta(b^\rho, a^\rho))}{2}-\frac{\rho^{\alpha}\Gamma(\alpha+1)}{2\eta(b^\rho,a^\rho)^{\alpha}}\bigg[{}^\rho I_{\left(\sqrt[\rho]{a^\rho+\eta(b^\rho,a^\rho)}\right)-}^\alpha f(a^\rho)\\
\nonumber
&\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad+{}^\rho I_{a+}^\alpha f(a^\rho+\eta(b^\rho,a^\rho))\bigg]\bigg|\\
&\leq\frac{\eta(b^\rho,a^\rho)}{\alpha+1}\left(1-\frac{1}{2^\alpha}\right)\left(\max\Big\{|f'(a^\rho)|^q, |f'(b^\rho)|^q\Big)\Big\}\right)^{1/q}.
\end{align}
\end{theorem}
\begin{proof}
Using Lemma \ref{mlem2}, the power mean inequality and the prequasiinvexity of $|f'|^q$, we have
\begin{align}\label{E0}
\nonumber
&\bigg|\frac{f( a^\rho)+f( a^\rho+\eta(b^\rho, a^\rho))}{2}-\frac{\rho^{\alpha}\Gamma(\alpha+1)}{2\eta(b^\rho,a^\rho)^{\alpha}}\bigg[{}^\rho I_{\left(\sqrt[\rho]{a^\rho+\eta(b^\rho,a^\rho)}\right)-}^\alpha f(a^\rho)\\
\nonumber
&\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad+{}^\rho I_{a+}^\alpha f(a^\rho+\eta(b^\rho,a^\rho))\bigg]\bigg|\\
\nonumber
&\leq\frac{\eta(b^\rho,a^\rho)\rho}{2}\left(\int_0^1\Big|(1-t^\rho)^\alpha-t^{\rho\alpha}\Big|t^{\rho-1}dt\right)^{1-1/q}\\
\nonumber
&\quad\quad\times\left(\int_0^1\Big|(1-t^\rho)^\alpha-t^{\rho\alpha}\Big|t^{\rho-1}\Big|f'( a^\rho+(1-t^\rho)\eta(b^\rho, a^\rho))\Big|^qdt\right)^{1/q}\\
\nonumber
&\leq\frac{\eta(b^\rho,a^\rho)\rho}{2}\left(\int_0^1\Big|(1-t^\rho)^\alpha-t^{\rho\alpha}\Big|t^{\rho-1}dt\right)\left(\max\Big\{|f'(a^\rho)|^q, |f'(b^\rho)|^q\Big\}\right)^{1/q}\\
&=\frac{\eta(b^\rho,a^\rho)\rho}{2}\left(\frac{1}{\rho}\int_0^1\Big|(1-u)^\alpha-u^{\alpha}\Big|du\right)\left(\max\Big\{|f'(a^\rho)|^q, |f'(b^\rho)|^q\Big\}\right)^{1/q}.
\end{align}
Now, we observe that
\begin{align}\label{E1}
\nonumber
\int_0^1\Big|(1-u)^\alpha-u^{\alpha}\Big|du&=\int_0^{1/2}\Big((1-u)^\alpha-u^{\alpha}\Big)du+\int_{1/2}^1\Big(u^\alpha-(1-u)^\alpha\Big)du\\
\nonumber
&=\frac{1}{\alpha+1}-\frac{1}{2^\alpha(\alpha+1)}+\frac{1}{\alpha+1}-\frac{1}{2^\alpha(\alpha+1)}\\
&=\frac{2}{\alpha+1}\left(1-\frac{1}{2^\alpha}\right).
\end{align}
The inequality in \eqref{MR3ineq} follows from \eqref{E0} and \eqref{E1}.
\end{proof}
\begin{remark}
If $\eta(x, y)=x-y$ in Theorem \ref{MR3}, then we recover the result in \cite[Theorem 2.4]{SM}. Also, if $\rho=1$ in Theorem \ref{MR3}, then we obtain the result in \cite[Theorem 2.3]{II}.
\end{remark}
%%%%%%%%%%%%%%%MR4
\begin{theorem}\label{MR4}
Under the conditions of Lemma \ref{mlem2}, if $|f'|^q, q>1$ is prequasiinvex on $I$, then the following inequality holds:
\begin{align}\label{MR4ineq}
\nonumber
&\bigg|\frac{f( a^\rho)+f( a^\rho+\eta(b^\rho, a^\rho))}{2}-\frac{\rho^{\alpha}\Gamma(\alpha+1)}{2\eta(b^\rho,a^\rho)^{\alpha}}\bigg[{}^\rho I_{\left(\sqrt[\rho]{a^\rho+\eta(b^\rho,a^\rho)}\right)-}^\alpha f(a^\rho)\\
\nonumber
&\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad+{}^\rho I_{a+}^\alpha f(a^\rho+\eta(b^\rho,a^\rho))\bigg]\bigg|\\
&\leq\frac{\eta(b^\rho,a^\rho)}{2}\left(\int_0^1\Big|(1-u)^\alpha-u^{\alpha}\Big|^rdu\right)^{1/r}\left(\max\Big\{|f'(a^\rho)|^q, |f'(b^\rho)|^q\Big\}\right)^{1/q},
\end{align}
where $\displaystyle\frac{1}{r}+\frac{1}{q}=1$. In addition, if $\alpha\in(0, 1]$, then we have the inequality
\begin{align}\label{MR4ineq2}
\nonumber
&\bigg|\frac{f( a^\rho)+f( a^\rho+\eta(b^\rho, a^\rho))}{2}-\frac{\rho^{\alpha}\Gamma(\alpha+1)}{2\eta(b^\rho,a^\rho)^{\alpha}}\bigg[{}^\rho I_{\left(\sqrt[\rho]{a^\rho+\eta(b^\rho,a^\rho)}\right)-}^\alpha f(a^\rho)\\
\nonumber
&\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad+{}^\rho I_{a+}^\alpha f(a^\rho+\eta(b^\rho,a^\rho))\bigg]\bigg|\\
&\leq\frac{\eta(b^\rho,a^\rho)}{2}\left(\frac{1}{\alpha r+1}\right)^{1/r}\left(\max\Big\{|f'(a^\rho)|^q, |f'(b^\rho)|^q\Big\}\right)^{1/q}.
\end{align}

\end{theorem}
\begin{proof}
Using Lemma \ref{mlem2}, the H\"older's inequality and the prequasiinvexity of $|f'|^q$, we have
\begin{align}
\nonumber
&\bigg|\frac{f( a^\rho)+f( a^\rho+\eta(b^\rho, a^\rho))}{2}-\frac{\rho^{\alpha}\Gamma(\alpha+1)}{2\eta(b^\rho,a^\rho)^{\alpha}}\bigg[{}^\rho I_{\left(\sqrt[\rho]{a^\rho+\eta(b^\rho,a^\rho)}\right)-}^\alpha f(a^\rho)\\
\nonumber
&\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad\quad+{}^\rho I_{a+}^\alpha f(a^\rho+\eta(b^\rho,a^\rho))\bigg]\bigg|\\
\nonumber
&\leq\frac{\eta(b^\rho,a^\rho)\rho}{2}\left(\int_0^1\Big|(1-t^\rho)^\alpha-t^{\rho\alpha}\Big|^rt^{\rho-1}dt\right)^{1/r}\\
\nonumber
&\quad\quad\times\left(\int_0^1t^{\rho-1}\Big|f'( a^\rho+(1-t^\rho)\eta(b^\rho, a^\rho))\Big|^qdt\right)^{1/q}\\
\nonumber
&\leq\frac{\eta(b^\rho,a^\rho)\rho}{2}\left(\frac{1}{\rho}\int_0^1\Big|(1-u)^\alpha-u^{\alpha}\Big|^r
du\right)^{1/r}\left(\frac{1}{\rho}\max\Big\{|f'(a^\rho)|^q, |f'(b^\rho)|^q\Big\}\right)^{1/q}\\
\nonumber
&=\frac{\eta(b^\rho,a^\rho)}{2}\left(\int_0^1\Big|(1-u)^\alpha-u^{\alpha}\Big|^rdu\right)^{1/r}\left(\max\Big\{|f'(a^\rho)|^q, |f'(b^\rho)|^q\Big\}\right)^{1/q}.
\end{align}
This proves the inequality in \eqref{MR4ineq}. By using Lemma \ref{SO}  with $\alpha\in(0, 1]$, we deduce that
\begin{align}\label{F1}
\nonumber
\int_0^1\Big|(1-u)^\alpha-u^{\alpha}\Big|^rdu&\leq\int_0^1\Big|1-2u\Big|^{\alpha r}du\\
\nonumber
&=\int_0^{1/2}(1-2u)^{\alpha r}du+\int_{1/2}^1(2u-1)^{\alpha r}du\\
\nonumber
&=\frac{1}{2(\alpha r+1)}+\frac{1}{2(\alpha r+1)}\\
&=\frac{1}{\alpha r+1}.
\end{align}
The inequality in \eqref{MR4ineq2} follows from \eqref{MR4ineq} and \eqref{F1}.
\end{proof}
\begin{remark}
If $\rho=1$ in the inequality \eqref{MR4ineq2} in Theorem \ref{MR4}, then we obtain the result in \cite[Theorem 2.4]{II}.
\end{remark}
%%%%%%%%corthm4
\begin{corollary}
If in Theorem \ref{MR4} we take $\eta(x, y)=x-y$ for all $x, y\in I$, i.e, $|f'|^q, q>1$, is quasiconvex, then the following inequality holds:
\begin{align}
\nonumber
&\bigg|\frac{f( a^\rho)+f(b^\rho)}{2}-\frac{\rho^{\alpha}\Gamma(\alpha+1)}{2(b^\rho-a^\rho)^{\alpha}}\bigg[{}^\rho I_{b-}^\alpha f(a^\rho)+{}^\rho I_{a+}^\alpha f(b^\rho)\bigg]\bigg|\\
\nonumber
&\leq\frac{b^\rho-a^\rho}{2}\left(\int_0^1\Big|(1-u)^\alpha-u^{\alpha}\Big|^rdu\right)^{1/r}\left(\max\Big\{|f'(a^\rho)|^q, |f'(b^\rho)|^q\Big\}\right)^{1/q},
\end{align}
where $\displaystyle\frac{1}{r}+\frac{1}{q}=1$. In addition, if $\alpha\in(0, 1]$, then we have the inequality
\begin{align}
\nonumber
&\bigg|\frac{f( a^\rho)+f(b^\rho)}{2}-\frac{\rho^{\alpha}\Gamma(\alpha+1)}{2(b^\rho-a^\rho)^{\alpha}}\bigg[{}^\rho I_{b-}^\alpha f(a^\rho)+{}^\rho I_{a+}^\alpha f(a^\rho+\eta(b^\rho,a^\rho))\bigg]\bigg|\\
\nonumber
&\leq\frac{b^\rho-a^\rho}{2}\left(\frac{1}{\alpha r+1}\right)^{1/r}\left(\max\Big\{|f'(a^\rho)|^q, |f'(b^\rho)|^q\Big\}\right)^{1/q}.
\end{align}

\end{corollary}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%conclusion
\section{Conclusion}
We established two midpoint-type inequalities and two trapezoidal-type inequalities for functions whose derivatives in absolute value to some powers are prequasiinvex with respect to a bifunction $\eta$ via the Katugampola fractional integral operators. By considering the bifunction $\eta(x, y)=x-y$,  the results for quasiconvex functions has been obtained from our main results. Several other results can be obtained from our results by considering different bifunctions and/or different values of the parameters involved. In particular, if we take $\rho=1$, then our results are in terms of the Riemann--Liouville fractional integrals. Also, we hope that under certain conditions on $f$ and $\eta$, similar results via the Hadamard fractional integrals could be derived from our results by taking the limit as $\rho\to0^+$. The details are left for the interested reader.

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%references



\end{document}
